

Temi di Discussione

(Working Papers)

Estimating the money market microstructure with negative and zero interest rates

by Edoardo Rainone and Francesco Vacirca







Temi di discussione

(Working papers)

Estimating the money market microstructure with negative and zero interest rates

by Edoardo Rainone and Francesco Vacirca

Number 1059 - February 2016

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Pietro Tommasino, Piergiorgio Alessandri, Valentina Aprigliano, Nicola Branzoli, Ines Buono, Lorenzo Burlon, Francesco Caprioli, Marco Casiraghi, Giuseppe Ilardi, Francesco Manaresi, Elisabetta Olivieri, Lucia Paola Maria Rizzica, Laura Sigalotti, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

ESTIMATING THE MONEY MARKET MICROSTRUCTURE WITH NEGATIVE AND ZERO INTEREST RATES

by Edoardo Rainone* and Francesco Vacirca[#]

Abstract

Money market microstructure is fundamental to studying bank behaviour, to evaluating monetary policy and to assessing the financial stability of the system. Given the lack of granular data on interbank loans, Furfine (1999) proposed an algorithm to estimate the microstructure using data from the payment system. We propose an econometric methodology to assess and improve the quality of the money market microstructure estimated by the Furfine algorithm in the presence of zero and negative rates, exploiting information coming from market regularities. We first extend the standard Furfine algorithm to include negative rates and verify the presence of significant noise at a specific rate. Secondly, we propose an inferential procedure that enriches and corrects the standard algorithm based on the economic likelihood of loans. Market regularities observed in this decentralized market are used to increase the reliability of the estimated interbank network. Thirdly, the methodology is applied to TARGET2, the European wholesale payment system. The main impacts of recent monetary policy decisions on key interest rates are studied, comparing the standard algorithm with the new econometric procedure.

JEL Classification: E52, E40, C21, G21, D40.

Keywords: interbank markets, money, payment systems, trading networks, measurement error.

Contents

1. Introduction	5
2. Preliminary evidence: the Furfine algorithm with negative and zero interest rates	7
3. The econometric methodology	8
4. Empirical analysis	12
5. Robustness checks	13
6. Concluding remarks and future research	14
References	16
Appendix	18
Tables	19
Figures	22

^{*}Bank of Italy, Directorate General for Markets and Payment Systems. Correspondent author: edoardo.rainone@bancaditalia.it

^{\$} ECB, DG Market Infrastrucure and Payments and Bank of Italy, Directorate General for Markets and Payment Systems.

1 Introduction

Knowledge of the interbank money market microstructure represents an important tool for monitoring and studying bank behaviour and for assessing monetary policy, both from a central bank and an academic perspective. In particular, it is helpful to evaluate the smoothness of the pass through mechanism and the financial stability of the system. These are the reasons why monetary policy makers, regulators, market operators and researchers closely look at money market indicators.

Several important papers used the interbank money market microstructure to study the financial and economic behaviour of banks. Among others, Ashcraft and Duffie (2007) showed how the intra-day allocation and pricing of overnight loans of federal funds reflect the decentralized inter-bank market in which these loans are traded. Afonso et al. (2011) examined the importance of liquidity hoarding and counterparty risk in the U.S. overnight interbank market during the financial crisis of 2008. Bech and Klee (2011) showed how successful the Federal Reserve was in raising the federal funds rate even in an environment with substantial reserve balances. Furfine (2003) examined the degree to which the failure of one bank would cause the subsequent collapse of other banks. Acharya and Merrouche (2012) studied the liquidity demand of large settlement banks in the UK and its effect on the money markets before and during the subprime crisis of 2007-08.

Given the lack of granular data about interbank loans, several approaches have been proposed to estimate the bilateral exposures in the money market. One body of literature drew inference from the bilateral exposures by starting with aggregate interbank assets and liabilities in bank balance sheet and applying information theory. Among others, Sheldon and Maurer (1998) and Helmut et al. (2013) applied the maximum entropy method, while Anand et al. (2014) used a minimal density approach. Another method for identifying the money market microstructure uses payment system data. Given that central bank money is exchanged in order to meet the reserve requirement and to settle payments, loans are usually settled in the payment system owned by the central bank in advanced economies.¹ Indeed, a loan is composed of two payments by definition, the payment corresponding to the setup of the loan, say from bank A to bank B, of an amount equal to p_t at day t and the reimbursement payment at day t + k of an amount equal to $p_{t+k} = p_t(1+i)$ from bank B to bank A, where k is the duration of the loan and i the interest rate. Exploiting this information, Furfine (1999) proposed an algorithm that attempts to match payments from the same money market contract. The original algorithm was designed to identify overnight transactions within Fedwire payment system data, assuming a minimum contract value of \$1 million and an interest rate within a corridor of 100 basis points centered around the federal funds rate. Observe that almost all the papers mentioned above, which study the behaviour of banks within the interbank money market microstructure, used data generated by this algorithm. Several authors have applied Furfine's method to data from different payment systems and have proposed several improvements to increase the scope and the quality of the estimation. Demiral et al. (2004) modified the algorithm to include loans of lower amounts and to limit the possible interest rates to

¹Many types of transactions are settled within the payment system. Here is a short list: customer payments, securities systems payments, open market operations, treasury bond issues. This should give an idea of the importance of the payment system and its centrality to banks from a liquidity management perspective. Reserve requirements oblige banks to hold a certain average amount of central bank money in their accounts during a maintenance period. A maintenance period is a time interval during which the amount of central bank money is averaged.

1/32 of a percentage point or use integer values to replicate the convention of the US government securities market. This limitation aims to decrease the number of false transactions detected. Akram and Christophersen (2010) implemented an algorithm for the Norwegian market, whereas Hendry and Kamhi (2007) applied it to the Canadian Large Value Transfer System (LVTS) by excluding all matches between payments whose estimated interest rates did not correspond to rates expressed in units of half a basis point. Arciero et al. (2013) extended the search algorithm to maturities of up to one year and applied the algorithm to payments settled through TARGET2 using different corridors;² the results obtained were validated using EONIA panel data and loan data from the e-MID money market trade platform. Recently some works have focused on assessing the quality of original Furfine-based algorithms (see for instance Armantier and Copeland (2012) and Kovner and David (2013)).

All the main implementations of the Furfine algorithm implicitly assume that the key interest rates are strictly positive. To the best of our knowledge no work looks at the application of the algorithm when interest rates may be negative or equal to zero. This issue has gained relevancy as a result of recent decisions taken by several monetary policy authorities to decrease the lower bound of the interest rate corridor, first to zero and then to negative values (e.g. in the euro area and in countries such as Switzerland, Sweden and Denmark). In particular in the euro area, to provide incentives to banks to trade more following the interbank freeze triggered by the sovereign crises in 2011, the Governing Council of the European Central Bank set the overnight deposit (OD) rate equal to zero in June 2012, lowered it to -0.1% as from June 2014 and recently, from September 2014, to -0.2%.

The main goal of this paper is to extend the Furfine algorithm to correctly identify money market transactions exchanged at zero or negative rates.

As a first step, we adapt the Arciero et al. (2013) implementation of the algorithm to include negative rates or equal to zero and we evaluate its robustness. The analysis of the results shows that the microstructure obtained is unreliable: by applying a formal statistical test to the output of the algorithm, it is shown that the algorithm fails to identify zero-rate transactions. It turns out that the standard version of the Furfine algorithm is not able to detect loans traded at a zero rate with a good degree of reliability.

As a second step, to overcome the identified issue concerning zero rates, we propose an econometric methodology that takes advantage of the information on regularities that are observed in a *decentralized market*, where loans' rates are agreed bilaterally by the counterparties. This method robustly estimates the market microstructure even when rates may be equal to zero. Grounded in the economic theory of over-the-counter markets, see Afonso and Lagos (2012) and Babus and Kondor (2013) among others, and based on the empirical literature, see Ashcraft and Duffie (2007), Afonso et al. (2011) and Angelini et al. (2011) for example, we use information on market regularities observed in the bilateral rate formation process to robustly estimate market microstructure. The underlying idea is to use economic and econometric theory to correctly detect loans by exploiting the information on market regularities, to estimate the likelihood of observing a false loan and, finally, to correct the initial microstructure through additional steps. We show its practical benefits when applied to the zero interest rate. Nevertheless, the usefulness of the methodology is broader: if applied to the full range of observed interest rates, it can decrease the generalized likelihood of detecting

²TARGET2 is the European real-time gross settlement payment system (RTGS) and the platform where the reserve requirement is managed, payments are settled and central bank money can be traded by banks. For more information about TARGET2 see http://www.ecb.europa.eu/paym/t2/html/index.en.html.

false loans at every rate.

As a third step, we exploit the results obtained in order to describe the main impact of recent variations in the Eurosystem's key interest rates on the money market microstructure. When we compare the results of a standard implementation of the Furfine algorithm that includes negative rates with the filtered results of the proposed methodology, we see that the effects of introducing a negative OD rate is quite different when our methodology is used. The increase in the volume and the value of loans is less evident and the relative weight of negative rates is higher.

The rest of the paper is organized as follows. Section 2 describes the results of the standard implementation of the Furfine algorithm using TARGET2 data when interest rates may be negative or equal to zero. Section 3 outlines the econometric methodology proposed in the paper. Section 4 presents the results of the application of our methodology using TARGET2 data. Section 5 presents the robustness checks and Section 6 the conclusions.

2 Preliminary Evidence: The Furfine Algorithm with Negative and Zero Interest Rates

We start with the implementation of the Furfine algorithm described in Arciero et al. (2013). The main features of the algorithm proposed in that paper are: i) it is designed to identify money market transactions with maturities up to one year, ii) the implied interest rate is limited to a corridor between r_{min} and r_{max} and must be a multiple of half a basis point, and iii) in case of multiple matches, the algorithm includes a procedure for run-time selection of the most plausible match. In this work, the implementation characteristics include: the selection of a 200 basis point corridor width³, between r_{min} and r_{max} , centred around the EONIA rate, the modification of the algorithm to include rates that are multiples of 0.1 basis points, thereby aligning it with the minimum rate tick used in euro money market (see, for instance, the e-MID platform rate tick); the application of the algorithm to ordering and beneficiary institution information drawn from payment system data, focusing on settled overnight transactions only.

To deal with the matter of negative rates, we removed the condition that, in the event the overnight deposit facility rate r_{OD} (OD rate) is lower than zero, the amount of the reimbursement transaction at day t + k, must be greater than the amount of the setup transaction at day t. That is, if the rate r_{OD} is greater than or equal to zero, the Furfine algorithm matches payments under the constraint that $p_t < p_{t+k}$, where p_t is the loan at time t and p_{t+k} is the repayment at time t + k, where k is the maturity. When $r_{OD} < 0$ the condition is removed and the amount of the reimbursement transaction, p_{t+k} , can be lower than, equal to or greater than the amount of the setup transaction, p_t , implicitly assuming that the contracted interest rate can be lower than, equal to or greater than 0 respectively.

In this section we report some of the aggregate results from the standard implementation of the algorithm, only removing the constraint, i.e. we assume it plausible to observe loans at zero or negative interest rates from June 2014, when the ECB Governing Council set the overnight deposit facility rate at -0.1%. From Figure 1 it appears that the daily average

 $^{^{3}}$ The same methodology described in the next sections was applied to the output of the algorithm using the other corridors presented in Arciero et al. (2013); the results obtained , not presented in this paper, show that the methodology is not impacted by the choice of corridor.

rate computed on estimated loans (FEONIA) is pretty close to the EONIA even after June 2014. If we look at panel (a) of Figure 2, which reports the daily number of overnight loans detected by the algorithm, it seems that the Eurosystem's decision increased market thickness. The same observation can be made for panel (b), where the daily value of those loans is reported.⁴ Note that these results lead to the conclusion that the Eurosystem's decision favoured a significant increase in interbank loans. Nevertheless, we may think that the strong discontinuity observed in June 2014 may be due to the inclusion of zero and negative interest rates. In particular, it seems that daily volatility strongly increased too. Observe that high volatility could characterize financial crises (see Figure 2). Given that no crisis was observed in June 2014, this evidence may indicate the presence of noise introduced by false detection of loans.⁵ If we focus on the time interval in which the OD is less than zero (from June 2014), we notice that many loans are agreed at negative rates, but the majority are settled at a zero rate (Figure 3). The graphical analysis of the empirical PDF and CDF of the rates shows an extremely high concentration of loans at zero rate (over 40% in both of the time intervals considered in Figure 3). This evidence supports the idea that some payments with the same value are paired by the algorithm even though they do not constitute a loan and are thus falsely labelled as money market transactions. The doubt is supported by the low degree of similarity between the CDFs (and PDFs) of FEONIA and EONIA (Figure 4). Apart from some discrepancies arising because they are computed using different samples, the two indices should be quite close to each other. These daily indicators are thus helpful in assessing the quality of the algorithm results. The possibility that ambiguous conclusions can be drawn about relevant central bank decisions means this issue must be addressed, using an accurate and formal approach.

3 The Econometric Methodology

Intuition. It therefore seems that the standard algorithm overestimates the number of loans at zero interest rate, but it is not clear a priori whether, and to what extent, this observation is correct. On the one hand, (i) it makes sense to lend money in the interbank market at a zero interest rate if the OD rate is less than zero. On the other hand, (ii) it is possible that pairs of payments that are not part of a loan agreement are imputed as loans. The difficulty lies in distinguishing the former case from the latter. It is an inferential issue, that stems from the fact that a loan can be observed and be true, as in the first case, or observed but instead be false, as in the second case.⁶ The economic theory suggests the existence of regularities in a decentralized market, where agents bilaterally agree on the price of the exchanged asset, see Afonso and Lagos (2012) and Babus and Kondor (2013) among others. It may help to assess the incidence of false loans and to identify them. In these markets, rate dispersion depends

⁴The strong decrease observed in July 2012, the date on which the OD was set to zero, might derive from the missing zero interest rate in the set of reasonable rates. There is an economic rationale for not including it; lending money or not lending money are equivalent choices from July 2012 to June 2014. Since lending requires at least the effort to find a counterparty, it does not seem useful to include the zero interest rate from July 2012. We delve further into this issue in Section 5.

⁵In addition, note that it can generate false signals to analysts monitoring rate volatility.

⁶From a financial logic perspective, the distinction between a zero interest rate loan and a pair of payments for the same amount lies in the different natures of the trades. In particular, if a payment does not represent either the supply or the repayment of a loan, a different *economic aspect* apply to this transaction.

on the characteristics of the counterparties, the date and the maturity:

$$r_{bl,m,t,n} = f(b, l, t, m, \bar{\theta}, \epsilon),$$

where $r_{bl,m,t,n}$ is the rate of the n_{th} loan at time t with maturity m having b as a borrower and l as lender, $\bar{\theta}$ is a set of parameters and ϵ is a random component. Intuitively, we can exploit observed regularities and estimate the likelihood of false loans for each point in the rate space. Examples of such regularities are the following:

- the rate paid during a day in which the amount of central bank money in the market is low is going to be higher than one in which the amount of central bank money is high;
- a risky borrower is going to pay a higher rate than a solid one;
- a lender with a structural excess of liquidity in its account has a propensity for lending at a lower rate compared with a lender with a low liquidity surplus;
- the rate paid for a long maturity is going to be higher than that for a short maturity;
- the rate paid reflects the bargaining powers of counterparties.

All these examples hold on average and all other things being equal. We can use an econometric model to embed these regularities, infer false loans by using the estimates obtained and precisely detect loans that are most likely false.

Assumptions. In order to keep things simple, we assume linearity, normally distributed unobservables and standard conditions for $f(\cdot)$.

Assumption 1. Let that the data generation process of the loan rate be the following

$$r_{bl,m,t,n} = \begin{cases} 1 - P(r_{bl,m,t,n} = r^{0}) & r^{0} \neq r_{bl,m,t,n} = \begin{cases} r_{bl,m,t,n} = \alpha_{t} + \beta_{b} + \gamma_{l} + \delta_{m} + \epsilon_{bl,m,t,n} \\ \epsilon_{bl,m,t,n} \sim N(0;\sigma_{t}) \end{cases}$$

$$r_{bl,m,t,n} = r^{0}) & r_{bl,m,t,n} = r^{0} = \begin{cases} P^{T} \begin{cases} r_{bl,m,t,n} = \alpha_{t} + \beta_{b} + \gamma_{l} + \delta_{m} + \epsilon_{bl,m,t,n} \\ \epsilon_{bl,m,t,n} \sim N(0;\sigma_{t}) \end{cases}$$

$$P^{F} \begin{cases} r_{bl,m,t,n} = \alpha_{t} + \beta_{b} + \gamma_{l} + \delta_{m} + \epsilon_{bl,m,t,n} \\ \epsilon_{bl,m,t,n} = \alpha_{t} + \beta_{b} + \gamma_{l} + \delta_{m} + \epsilon_{bl,m,t,n} \end{cases}$$

$$(1)$$

$$\sigma_t = s(|r_{ML,t} - r_{OD,t}|)$$

where $r_{bl,m,t,n}$ is the rate of the n_{th} loan at time t with maturity m with b as borrower and l as lender, α_t is the time fixed effect, β_b is the borrower fixed effect, γ_l is the lender fixed effect, δ_m is the maturity fixed effect and $\epsilon_{bl,m,t,n}$ is the normally distributed error. $r_{ML,t}$ is the marginal lending rate, $r_{OD,t}$ is the overnight deposit rate and $s(\cdot)$ is a function capturing the corridor width. P^T is the probability of observing a rate equal to r^0 when it is "true", while P^F is the probability of observing a rate equal to r^0 when it is "false". It implies that if the loan is false, the measurement error takes an explicit form, as reported in model (1). Observe that $P(r_{bl,m,t,n} = r^0) = P^T + P^F$. We allow for heteroskedasticity and let the error variance depend on the corridor width at time t with a functional form $s(\cdot)$.⁷

Assumption 2. The market rate is able to capture the daily liquidity conditions and it is independent to $\epsilon_{bl,m,t,n}$.

Estimated parameter properties under false loans. Under Assumption 2 we let the market rate, like the EONIA for euro funds, capture the daily conditions and thus, we can consider the following matrix form for model (1)

$$R = E\alpha + B\bar{\beta} + L\bar{\gamma} + M\bar{\delta} + \Sigma = X\bar{\theta} + \Sigma,$$

where β is a vector of borrower fixed effects with size equal to the number of borrowers in the market, $\bar{\gamma}$ is a vector of lender fixed effects with size equal to the number of lenders in the market. $\bar{\delta}$ is a vector of maturities fixed effects. E is a vector with a value equal to the market rate, E_k if t = k, while B, L, M are matrices of zeros and ones that keep track of the loan borrower, lender and maturity and associate them with the respective parameters included in $\bar{\beta}, \bar{\gamma}$ and $\bar{\delta}$. Σ is a vector of errors. X = [E, B, L, M] is the $n \times K$ matrix of regressors and $\bar{\theta'} = [\alpha, \bar{\beta'}, \bar{\gamma'}, \bar{\delta'}]'$. Splitting the sample based on which loans are false and which are true, we obtain

$$\begin{bmatrix} R^T \\ R^F \end{bmatrix} = \begin{bmatrix} E^T \\ E^F \end{bmatrix} \alpha + \begin{bmatrix} B^T \\ B^F \end{bmatrix} \bar{\beta} + \begin{bmatrix} L^T \\ L^F \end{bmatrix} \bar{\gamma} + \begin{bmatrix} M^T \\ M^F \end{bmatrix} \bar{\delta} + \begin{bmatrix} \Sigma^T \\ \Sigma^F \end{bmatrix} = \begin{bmatrix} X^T \\ X^F \end{bmatrix} \bar{\theta} + \begin{bmatrix} \Sigma^T \\ \Sigma^F \end{bmatrix},$$

where the size of T elements is $N(1 - P^F)$, while the size of F elements is NP^F . Under Assumptions 1 - 2 where we have $X^{T'}\Sigma^T = 0_K$ and where 0_K is a $1 \times K$ vector of zeros, it turns out that

$$E(X'\Sigma) = f(P^F) = \begin{cases} 0 & P^F = 0\\ -X^{F'}X^F\bar{\theta} & P^F = 1 \end{cases}$$

In other words, when there are no false loans in the sample, the errors are not correlated with the regressors, while this correlation increases with the percentage of false loans. Consequently the following limits can be derived for the OLS estimates of the model parameters,⁸

$$\lim_{P^F \to 0} \hat{\alpha}_{OLS} = \alpha,$$

⁸Observe that if we assume that the covariance between the regressors is zero we also have:

$$\begin{split} E(E'\Sigma) &= f(P^F) = \left\{ \begin{array}{cc} 0 & P^F = 0 \\ -E^{F'}E^F\alpha & P^F = 1 \end{array} \right., \\ E(B'\Sigma) &= f(P^F) = \left\{ \begin{array}{cc} 0 & P^F = 0 \\ -B^{F'}B^F\bar{\beta} & P^F = 1 \end{array} \right., \\ E(L'\Sigma) &= f(P^F) = \left\{ \begin{array}{cc} 0 & P^F = 0 \\ -L^{F'}L^F\bar{\gamma} & P^F = 1 \end{array} \right., \\ E(M'\Sigma) &= f(P^F) = \left\{ \begin{array}{cc} 0 & P^F = 0 \\ -M^{F'}M^F\bar{\delta} & P^F = 1 \end{array} \right., \end{split}$$

⁷We can allow σ_t to also depend on $r_{EONIA,t}$, the EONIA rate.

$$\lim_{P^F \to 0} \hat{\bar{\beta}}_{OLS} = \bar{\beta},$$
$$\lim_{P^F \to 0} \hat{\bar{\gamma}}_{OLS} = \bar{\gamma},$$
$$\lim_{P^F \to 0} \hat{\bar{\delta}}_{OLS} = \bar{\delta}.$$

Under Assumptions 1 and 2 the restricted sample on $r_{bl,m,t,n} \neq r_0$ does not contain false loans, while the unrestricted sample may. It follows that a set of formal tests for the significant presence of false rates equal to r^0 is

$$H_0 : \hat{\alpha}_{OLS,R} - \hat{\alpha}_{OLS,U} = 0,$$

$$H_0 : \hat{\bar{\beta}}_{OLS,R} - \hat{\bar{\beta}}_{OLS,U} = 0,$$

$$H_0 : \hat{\bar{\gamma}}_{OLS,R} - \hat{\bar{\gamma}}_{OLS,U} = 0,$$

$$H_0 : \hat{\bar{\delta}}_{OLS,R} - \hat{\bar{\delta}}_{OLS,U} = 0.$$

Where the subscript U defines the unrestricted sample, the subscript R defines the restricted subsample of $r_{bl,m,t,n} \neq r_0$. The test follows from the consistency of $\hat{\theta}_R$ and the possible inconsistency of $\hat{\theta}_U$ under Assumptions 1 and 2 (see the Appendix for more details). In other words, $\hat{\theta}_R$ provides us with robust information about market regularities, while the divergence of $\hat{\theta}_U$ signals the presence of false loans, that by definition do not follow such regularities. Let $D_{U,R} = \bar{\theta}_U - \bar{\theta}_R$ be the difference between the OLS estimators of the restricted and unrestricted sample. From the last equality of (5), in the Appendix, we can derive the expected value of the bias as a function of the probability of observing false rates:

$$E(D_{U,R}) = E\left((nVC(X))^{-1}[n_FVC(X_F); n_TC(X_T, \Sigma^T)][-\bar{\theta}'; 1]'\right) = [P_F I_K; (1 - P_F)VC(X)^{-1}C(X, \Sigma)][-\bar{\theta}'; 1]' = ([O_K; o_K] + P_F [I_K; o_K])[-\bar{\theta}'; 1]',$$
(2)

where VC(X) = (X'X)/n and $VC(X_F) = (X'_FX_F)/n_F$ are the variance-covariance matrices computed on different samples, I_K is the identity matrix with dimension K, O_K and o_K are respectively a square matrix and a vector of zeros with dimension K. $C(X_T, \Sigma^T) = (X'_T\Sigma_T)/n_T$ is the sample covariance matrix between the error term and the covariates, while $C(X, \Sigma)$ is the covariance matrix of the respective random variables. From the last equality in (2) it turns out that a consistent estimator for P^F is

$$\hat{P}^F = \hat{\theta}^+ \hat{D}_{U,R},\tag{3}$$

where $\hat{D}_{U,R}$ is a consistent estimator of the bias and $\hat{\theta}^+$ is the Moore-Penrose pseudoinverse of a consistent estimator of the true parameters. Under Assumptions 1 and 2 both can be recovered using OLS estimates of $\bar{\theta}_U$ and $\bar{\theta}_R$. The intuition is as follows: the restricted model (R) gives consistent estimates of the model parameters $(\bar{\theta})$, providing the econometrician with a correct characterization of market regularities, while the estimates from the unrestricted model (U), departing from the true value as a function of the likelihood of observing false loans at a rate equal to r_0 , make it possible to quantify the bias $(D_{U,R})$. A test for false loans. Having estimates from the restricted and unrestricted models allows us to test whether the presence of false loans is significant for a specific rate r_0 .

$$T = (\hat{\bar{\theta}}_U - \hat{\bar{\theta}}_R)' diag(1/\sqrt{\hat{\sigma}_{\bar{\theta}_U}^2 + \hat{\sigma}_{\bar{\theta}_R}^2}), \tag{4}$$

where $\hat{\sigma}_{\bar{\theta}_U}$ and $\hat{\sigma}_{\bar{\theta}_R}$ are the $K \times 1$ vectors of the standard deviations of the estimated parameters respectively from the unrestricted and the restricted models and $diag(\cdot)$ transforms the vector into a diagonal matrix. The vector T gives K p-values, as a standard test for significant difference with $\theta_U^k = \theta_R^k$, $k = 1, \dots, K$, as the null hypotheses; the greater the distance from 0.5 in absolute value, the greater the likelihood that false loans at rate r_0 are present.

A correction procedure for the market microstructure. If the test signals a high likelihood of false loans, we can detect the loans that are the most likely to be false using the following simple procedure. Take the vector $\hat{\epsilon}^U$, which contains the residuals from the unrestricted model and has $\hat{\epsilon}^U_{bl,t,m,n}$ as a generic element and the time-varying estimates of its variance $\hat{\sigma}^U_t$. Then, compute the $(1 - \hat{P}_F)_{th}$ percentile of its empirical distribution, $\tau^{1-\hat{P}_F}_{\hat{\epsilon}^U}$. The loans most likely to be false are contained in this set:

$$F(r_0) := \{ L_{bl,t,m,n} : (r_{bl,t,m,n} = r_0) \cap (\frac{\hat{\epsilon}_{bl,t,m,n}^U}{\hat{\sigma}_t^U} \ge \tau_{\hat{\epsilon}^U}^{1-\hat{P}_F}), \}$$

where $L_{bl,t,m,n}$ is the n_{th} loan at time t with maturity m having b as a borrower and l as lender. It is thus possible to clean our dataset of loans less likely to be true, identifying them precisely. In this way we can also study their characteristics ex-post in order to better identify the source of noise, using supplementary information if necessary.

4 Empirical Analysis

We implemented the procedure outlined in Section 3 for the set of loans detected by the standard Furfine algorithm applied to TARGET2 data.⁹ The parameter estimates for the unrestricted and restricted (with $r_0 = 0$) models are reported in Table 1 columns (1)-(2). Most of the estimated coefficients differ significantly across these two subsamples. Furthermore some of the fixed effects are significant in the restricted sample while not so in the unrestricted, and viceversa. The hint is supported by the test proposed in equation (4) and reported in column (1) of Table 2. The relative p-values reported in column (2) show a quite significant difference between the two samples. In particular, the time coefficient α is very different, indicating that in the unrestricted sample, the dependence on the day in which the loan is agreed is lower. In other words, it means that, all things being equal, the number of estimated loans at a zero interest rate is almost independent of the average market rate, i.e. very likely in the days on which the average market rate is far from zero.

Our estimate of P_F , using estimand (3), is equal to 0.38, roughly 80% of the detected loans at zero interest rate (which in turn are 48% of the total). We delete this percentage of

⁹The econometric procedure is applied to the time interval starting from 11 June 2014. The reasons behind it are at least two. The first one is theoretical, there is no incentive to trade at zero interest rate when the OD is equal or greater than zero. The second one is empirical, we do not observe zero interest rate loans traded on the e-MID platform before this date.

loans from the initial sample, selecting them by following the correction procedure outlined in Section 3.

These steps lead us to a new sample, from which the loans most likely to be false loans have been deleted. Let us describe this sample. The volume (panel (a)) and the value (panel (b)) are reported in Figure 5. We can see that the discontinuity (in correspondence to the implementation of the negative OD rate on 11 June 2014) observed in the initial sample vanishes. In terms of policy relevant facts, it is worthwhile to note that this new sample draws different conclusions, as compared with the initial sample, regarding the effects of the Eurosystem's decision. The volume decreases from 11 June 2014, instead of increasing as in the initial sample. The increase in value does not seem to be as prominent as before, showing a substantial stationarity. It implies an increase in the average value of a loan. The new rate's empirical CDF and PDF do not signal the huge probability mass on zero (Figure 6). Furthermore, the frequency of negative interest rates significantly increases, moving from 16%to 20% during the maintenance periods in which the OD rate is equal to -0.1% and from 27%to 43% during the maintenance periods in which the OD rate is equal to -0.2%. Figure 7 shows the empirical CDF and PDF of the new FEONIA and the EONIA for both of the aforementioned maintenance periods. The similarity between these two indicators is significantly improved as compared with the initial sample, as the closeness of those functions highlights. To make the statement more robust we perform a number of similarity tests -the Kolmogorov-Smirnov test among others- using this sample and the one detected using the standard Furfine algorithm (Figure 4). The results are reported in Table 3. We clearly see that the p-values of the similarity tests are always closer to 0.5 when our additional step is performed and that the null hypothesis of belonging to the same distribution is almost never rejected.

5 Robustness Checks

To give us more confidence in the results obtained using the proposed econometric procedure, we implement, in this section, four robustness checks.

The first exercise consists in changing the reference interest rate (r_0 in the econometric procedure) from 0% to 0.01% and to -0.01%. Those rates are plausible, as zero is, but we empirically did not find an abnormal frequency of loans agreed at that rate. Furthermore, the related payments cannot be confounded with other liquidity transfers for the same value but of a different nature. The restricted model outlined in Section 3 is thus estimated with $r_0 = 0.01\%$ and $r_0 = -0.01\%$. Columns (3)-(4) of Table 1 reports the results, the estimated coefficients are very close to the ones from the unrestricted model in column (1), we also do not find any control to be significant in one specification and not in the other. Columns (3)-(4) and columns (5)-(6) of Table 2 show that we never reject the hypothesis of having the same coefficient for the restricted and the unrestricted sample. All p-values are extremely close to 0.5, which signals that there is no need to apply our procedure to these rates. The evidence indicates that the standard Furfine algorithm does a good job in matching loans with rates equal to 0.01% and -0.01% and that our econometric procedure is valuable especially for filtering out false loans at zero interest rate.

In the second exercise we apply our econometric procedure to a different sample. The latter is generated by assuming that zero interest rates may be observed even from 11 July 2012, the date on which the OD rate had been set to zero by the Eurosystem.¹⁰ As we saw in Figure 2, the inclusion of zero rates seems to generate remarkable noise. The evidence is confirmed by Figure 8 where the standard Furfine algorithm is allowed to pick loans at zero rate from 11 July 2012; the additional noise between 11 July 2012 and 11 June 2014 is visible in both the panels. Data from the e-MID platform does not contain loans at zero rate in this time interval, suggesting that our econometric procedure should yield a subset of loans very similar to the baseline sample. In other words, it should reject almost every additional loan at zero rate detected by the algorithm in the time interval between 11 July 2012 and 11 June 2014. It is interesting to observe how the effects of monetary policy look quite different if we include zero rates from 11 July 2012. Looking at Figure 8 it seems that the number of loans increased after 11 July 2012 and decreased after 11 June 2014, exactly the opposite conclusions drawn in Figure 2. Results from our econometric procedure applied to this sample are reported in Figure 9, note that the volume and value of loans are almost coincident with those from the baseline sample (in Figure 5).

The third exercise consists in applying the econometric procedure to the set of loans agreed using the e-MID platform. The number of false loans should be very close to zero because all of these are actual loans. In order to more accurately asses the robustness of our procedure, we allow the standard Furfine algorithm to include zero rates from 11 July 2012 and from 11 June 2014. In Figure 10 the time series of the frequency of zero interest rates loans are reported for the original data and for the samples returned by the econometric procedure when zero rates are included from 11 July 2012 (filtered 1) and from 11 June 2014 (filtered 2). The three series are very close; this highlights the success of the procedure in holding the actual loans at a zero interest rate. Indeed, the number of deleted loans is negligible. In the fourth exercise we apply the procedure (with the same reference rate $r_0 = 0$) to the final sample we obtained in Section 4, which is already cleaned of loans that are most likely false. If the methodology is robust we should not reject the null hypothesis of an insignificant number of false loans at zero rate. The relative estimates, tests and p-values are reported in Table 4, and show that we never reject the hypothesis. Observe also that the ones in column (1) of Table 4 are thus the reliable estimates for the model parameters.

6 Concluding Remarks and Future Research

The Furfine algorithm is a useful tool for policy makers and researchers in monetary economics and central banking and therefore, its reliability is fundamental. Recently some central banks are allowing negative and zero interest rates, a notable example being the Eurosystem's decision of June 2014 to set the overnight deposit rate to -0.01%. In this paper we assess the robustness of the algorithm when rates are permitted to be negative or equal to zero.

Our first result is that the algorithm is not reliable in such a context; in particular, the level of noise for zero interest rate is huge. We formally prove this by proposing a statistical test for the presence of significant noise at a specific rate. It turns out that the microstructure generated by the algorithm is strongly biased. Specifically, many liquidity transfers having different economic natures are matched and erroneously labelled as loans.

¹⁰It is possible that loans were agreed at zero interest rate even from that date. This possibility is partially supported by the remarkable decreases shown in Figure 2 in correspondence to July 2012. Nevertheless, there are no direct costs for having an excessive amount of central bank money between July 2012 and June 2014.

As a second contribution, we propose an econometric methodology to robustly estimate the market microstructure even in the presence of strong noise, i.e. zero interest rates. Taking advantage of the recent theoretical and empirical literature on OTC markets, we built a procedure based on the regularities observed in this decentralized market. In practice, we enrich the algorithm with two additional steps, which respectively detect and treat the undesirable noise. These additional steps formally take into account the economic likelihood of loans. In other words, the procedure employs the information on market regularities, enriching the estimation with inputs from the observation of systematic patterns within the market.

Finally, the paper describes the main impact of recent monetary policy decisions regarding key interest rates. We use reliable data from the proposed procedure, and show how different they are with respect to the dataset produced by the standard algorithm. In particular, if we want to evaluate the consequences of a negative overnight deposit rate, the conclusions drawn are drastically different. The volume and value of loans are definitely lower and the frequency of loans at negative interest rates is absolutely higher as compared with the standard algorithm.

These discrepancies highlight the importance of making an appropriate inference when the money market microstructure is estimated and shed light on the key role that economic and econometric methods can play in this context.

In this paper, to keep things straight, we use a simple ex-post frequentist approach to correct the market microstructure for noise. A more effective and complex approach that estimates microstructure and market regularities jointly, using a Bayesian procedure, can be used. We are exploring this possibility and leave it for future research.

References

- Acharya, V. V. and Merrouche, O. (2012). Precautionary hoarding of liquidity and interbank markets: Evidence from the subprime crisis, *Review of Finance*.
- Afonso, G., Kovner, A. and Schoar, A. (2011). Stressed, not frozen: The federal funds market in the financial crisis, *The Journal of Finance* **66**(4): 1109–1139.
- Afonso, G. and Lagos, R. (2012). Trade dynamics in the market for federal funds, *Technical report*, Staff Report, Federal Reserve Bank of New York.
- Akram, F. A. and Christophersen, C. (2010). Interbank overnight rates gains from systemic importance, Norges Bank Working Paper 11.
- Anand, K., Craig, B. and von Peter, G. (2014). Filling in the blanks: Network structure and interbank contagion, *Discussion Papers 02/2014*, Deutsche Bundesbank, Research Centre.
- Angelini, P., Nobili, A. and Picillo, C. (2011). The interbank market after august 2007: what has changed, and why?, *Journal of Money*, *Credit and Banking* 43(5): 923–958.
- Arciero, L., Heijmans, R., Heuver, R., Massarenti, M., Picillo, C. and Vacirca, F. (2013). How to measure the unsecured money market? the eurosystem's implementation and validation using target2 data, DNB Working Papers 369.
- Armantier, O. and Copeland, A. M. (2012). Assessing the quality of Furfine-Based algorithms, Staff Report 575, Federal Reserve Bank of New York.
- Ashcraft, A. B. and Duffie, D. (2007). Systemic illiquidity in the federal funds market, The American economic review pp. 221–225.
- Babus, A. and Kondor, P. (2013). Trading and information diffusion in over-the-counter markets, *Technical report*, Working paper.
- Bech, M. L. and Klee, E. (2011). The mechanics of a graceful exit: Interest on reserves and segmentation in the federal funds market, *Journal of Monetary Economics* **58**(5): 415–431.
- Demiralp, S., Preslopsky, B. and Whitesell, W. (2004). Overnight interbank loans, *Manuscript* Board of Governors of the Federal Reserve.
- Furfine, C. (1999). The microstructure of the federal funds market, Financial Markets, Institutions & Instruments 8(5): 24–44.
- Furfine, C. H. (2003). Interbank exposures: Quantifying the risk of contagion, Journal of money, credit and banking pp. 111–128.
- Helmut, E., Alfred, L. and Martin, S. (2013). Network models and systemic risk assessment, in J.-P. Fouque and J. A. Langsam (eds), Handbook on Systemic Risk, Cambridge University Press.
- Hendry, S. and Kamhi, N. (2007). Uncollateralized overnight loans settled in LVTS, Bank of Canada Working Paper 07-11.

- Kovner, A. and David, S. (2013). Evaluating the quality of fed funds lending estimates produced from fedwire payments data, *Staff Report 629*, Federal Reserve Bank of New York.
- Sheldon, G. and Maurer, M. (1998). Interbank lending and systemic risk: An empirical analysis for switzerland, Swiss Journal of Economics and Statistics (SJES) 134(IV): 685–704.

Appendix

The OLS estimators for the restricted and unrestricted sample are

$$\begin{bmatrix} \hat{\alpha}_{OLS,R} \\ \bar{\beta}_{OLS,R} \\ \bar{\hat{\gamma}}_{OLS,R} \\ \bar{\hat{\delta}}_{OLS,R} \end{bmatrix} = \left(\begin{bmatrix} E_R \\ B_R \\ L_R \\ M_R \end{bmatrix}' \begin{bmatrix} E_R \\ B_R \\ L_R \\ M_R \end{bmatrix} \right)^{-1} \begin{bmatrix} E_R \\ B_R \\ L_R \\ M_R \end{bmatrix}' R_R,$$

$$\begin{bmatrix} \hat{\alpha}_{OLS,U} \\ \bar{\hat{\beta}}_{OLS,U} \\ \bar{\hat{\gamma}}_{OLS,U} \\ \bar{\hat{\delta}}_{OLS,U} \end{bmatrix} = \left(\begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix}' \begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix} \right)^{-1} \begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix}' R,$$

$$R,$$

$$R^A_{R^R} \end{bmatrix} = \alpha \begin{bmatrix} E^A \\ E^R \end{bmatrix} + \bar{\beta} \begin{bmatrix} B^A \\ B^R \end{bmatrix} + \bar{\gamma} \begin{bmatrix} L^A \\ L^R \end{bmatrix} + \bar{\delta} \begin{bmatrix} M^A \\ M^R \end{bmatrix} + \begin{bmatrix} \Sigma^A \\ \Sigma^R \end{bmatrix}$$

and

with

where the subscript U defines the unrestricted sample, the subscript R defines the restricted subsample of $r_{bl,m,t,n} = r_0$, while the subscript A identifies the rest of the sample. The size of Relements is $N[1 - (P^F + P^T)]$, while the size of A elements is $N[P^F + P^T]$. If $P^F \neq 0$, because of the presence of false loans, only the estimates from the restricted sample are consistent, while the estimates from the unrestricted are biased:

,

$$E \begin{pmatrix} \begin{bmatrix} \hat{\alpha}_{OLS,R} \\ \bar{\beta}_{OLS,R} \\ \bar{\gamma}_{OLS,R} \\ \bar{\delta}_{OLS,R} \end{bmatrix} = \begin{bmatrix} \alpha \\ \bar{\beta} \\ \bar{\gamma} \\ \bar{\delta} \end{bmatrix},$$

$$E \begin{pmatrix} \begin{bmatrix} \hat{\alpha}_{OLS,U} \\ \bar{\beta}_{OLS,U} \\ \bar{\gamma}_{OLS,U} \\ \bar{\delta}_{OLS,U} \end{bmatrix} = \begin{bmatrix} \alpha \\ \bar{\beta} \\ \bar{\gamma} \\ \bar{\delta} \end{bmatrix} + \begin{pmatrix} \begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix}' \begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix} \end{pmatrix}^{-1} \begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix}' \begin{bmatrix} \Sigma^F \\ \Sigma^T \end{bmatrix},$$

$$\begin{bmatrix} \Sigma^F \\ \Sigma^T \end{bmatrix} = \begin{bmatrix} [E^F, B^F, L^F, M^F]'[-\alpha, -\bar{\beta}, -\bar{\gamma}, -\bar{\delta}] + r^0 \iota^F \\ \Sigma^T \end{bmatrix}.$$

where

Let $r^0 = 0$, we then have

$$D_{U,R} = \bar{\theta}_{U} - \bar{\theta}_{R} = \left(\begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix}' \begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix} \right)^{-1} \left[\begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix}' \begin{bmatrix} [E^{F}, B^{F}, L^{F}, M^{F}]'[-\bar{\theta}] \\ \Sigma^{T} \end{bmatrix} \right]$$
$$= \left(\begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix}' \begin{bmatrix} E \\ B \\ L \\ M \end{bmatrix} \right)^{-1} \left\{ \begin{bmatrix} E^{F} \\ B^{F} \\ L^{F} \\ M^{F} \end{bmatrix}' \begin{bmatrix} E^{F} \\ B^{F} \\ L^{F} \\ M^{F} \end{bmatrix}' \begin{bmatrix} E^{F} \\ B^{F} \\ L^{F} \\ M^{F} \end{bmatrix}; \begin{bmatrix} E^{T} \\ B^{T} \\ L^{T} \\ M^{T} \end{bmatrix}' \Sigma^{T} \right\} \left\{ \begin{bmatrix} -\bar{\theta} \\ 1 \end{bmatrix} \right\}.$$
(5)

Tables

Depende	ent Variable: loa	n rate		
	Baseline	results	Robust	ness checks
	Unrestricted model	Restricted model	Restricted model	Restricted model
	(1)	$7 \neq 0$ (2)	$\begin{array}{c} 7 \neq 0.01 \\ (3) \end{array}$	$7 \neq -0.01$ (4)
α	0.4319^{***} (0.0052)	0.7124^{***} (0.0085)	0.4348^{***} (0.0054)	0.4317^{***} (0.0053)
ſ	0.0094***	0.0248***	0.0089***	0.0096***
	(0.0014)	(0.0022)	(0.0015)	(0.0014)
	-0.0020	(0.0180***	-0.0035	-0.0025
	0.0044***	0.0602***	0.0048***	0.0044***
	(0.0015)	(0.0065)	(0.0016)	(0.0016)
	0.0213***	0.0313^{***}	0.0222***	0.0215^{***}
	(0.0032)	(0.0059)	(0.0034)	(0.0033)
	(0.0143^{***})	(0.0442^{***})	(0.0146^{***})	(0.0145^{***})
	0.0381***	0.0667***	0.0431***	0.0382***
	(0.0019)	(0.0041)	(0.0021)	(0.0019)
	0.0043***	0.0265^{***}	0.0045^{***}	0.0045^{***}
	(0.0012)	(0.0024)	(0.0013)	(0.0013)
	-0.0020*	0.0072^{***}	-0.0016	-0.0020
	0.0083***	(0.0027) 0.0211***	0.0076***	0.0084***
$\beta_{\rm s}$	(0.0023)	(0.0039)	(0.0025)	(0.0023)
	0.0190***	0.0322***	0.0204***	0.0192***
	(0.0017)	(0.0034)	(0.0018)	(0.0017)
	-0.0019	0.0022	-0.0030	-0.0019
	(0.0024) 0.0005	(0.0041) 0.0150***	0.0026)	(0.0025) 0.0007
	(0.0015)	(0.0027)	(0.0016)	(0.0016)
	0.0099***	-0.0073	0.0106***	0.0100***
	(0.0018)	(0.0061)	(0.0019)	(0.0019)
	0.0342***	0.0358***	0.0380***	0.0359***
	-0.0234***	-0.0621***	-0.0200***	-0.0236***
	(0.0028)	(0.0049)	(0.0029)	(0.0029)
	0.0258 * * *	0.0378***	0.0288***	0.0260***
	(0.0032)	(0.0060)	(0.0035)	(0.0033)
	(0.0318^{***})	0.1074^{***}	(0.0330^{***})	(0.0320^{***})
	(0.0022)	(0.0000)	(0.0023)	(0.0022)
ſ	-0.0145***	-0.0334***	-0.0154***	-0.0143***
	(0.0014) 0.0208***	(0.0022) 0.0461***	(0.0015) 0.0214***	(0.0014) 0.0208***
	(0.0020)	(0.0031)	(0.0020)	(0.0020)
	-0.0083***	0.0120***	-0.0086***	-0.0083***
	(0.0016)	(0.0049)	(0.0016)	(0.0016)
	-0.0056	-0.0222***	-0.0058	-0.0055
	(0.0035)	(0.0061)	(0.0037)	(0.0035)
	(0.0011)	(0.0019)	(0.0011)	(0.0011)
	0.0116^{***}	0.0035	0.0134***	0.0120***
	(0.0018)	(0.0039)	(0.0020)	(0.0019)
	-0.0196***	-0.0451***	-0.0210***	-0.0196***
	-0.0201***	-0.0466***	-0.0206***	-0.0201***
	(0.0012)	(0.0028)	(0.0012)	(0.0012)
	-0.0133***	-0.0231***	-0.0146***	-0.0133***
$\gamma_{\rm s}$	(0.0026)	(0.0047)	(0.0028)	(0.0026)
<i>'</i>	(0.0078^{***})	(0.0056)	(0.0093^{***})	(0.0076^{***})
	-0.0097***	-0.0266***	-0.0099***	-0.0096***
	(0.0017)	(0.0033)	(0.0018)	(0.0017)
	-0.0078***	-0.0223***	-0.0085***	-0.0078***
	(0.0022)	(0.0036)	(0.0023)	(0.0022)
	-0.0288***	-0.0533***	-0.0295***	-0.0289***
	-0.0091***	0.0121	-0.0097***	-0.0092***
	(0.0019)	(0.0077)	(0.0020)	(0.0019)
	0.2502***	0.2854***	0.2739***	0.2486***
	(0.0034)	(0.0049) 0.1387***	(0.0037)	(0.0035)
	(0.0031)	(0.0059)	(0.0033)	(0.0032)
	0.0010	-0.0088	0.0011	0.0009
	(0.0031)	(0.0059)	(0.0033)	(0.0032)
	0.0391***	0.0378***	0.0400***	0.0390***
((0.0028)	(0.0067)	(0.0029)	(0.0028)

 Table 1: Rate Equation - Parameter Estimates

Notes: * : p < 0.10; **: p < 0.05; ***: p < 0.01. Only fixed effects with more than 1% of observations are included in the model. β s and γ s are country fixed effects for countries having a frequency higher than 0.01%.

Variab	ole: loan ra	ate				
	Baseline	e results		Robustne	ess checks	
	r =	= 0	r =	0.01	r = -	-0.01
	p(r) =	= 48%	p(r) =	= 7%	p(r) =	= 2%
	T (1)	$^{\mathrm{P}}_{(2)}$	Т (3)	P (4)	$^{\mathrm{T}}_{(5)}$	P (6)
α	-2.3956	0.0083	-0.0276	0.4890	0.0022	0.5009
(-0.2563	0.3989	0.0094	0.5037	-0.0039	0.4985
	-0.2322	0.4082	0.0123	0.5049	-0.0007	0.4997
	-0.6243	0.2662	-0.0067	0.4973	-0.0001	0.5000
	-0.1043	0.4585	-0.0113	0.4955	-0.0020	0.4992
	-0.5011	0.3081	-0.0055	0.4978	-0.0049	0.4981
	-0.3689	0.3561	-0.0786	0.4687	-0.0016	0.4994
	-0.3665	0.3570	-0.0031	0.4988	-0.0030	0.4988
	-0.1482	0.4411	-0.0076	0.4969	-0.0001	0.5000
βs	-0.1623	0.4355	0.0092	0.5037	-0.0019	0.4992
	-0.1848	0.4267	-0.0231	0.4908	-0.0033	0.4987
	-0.0513	0.4795	0.0148	0.5059	-0.0005	0.4998
	-0.2215	0.4123	-0.0054	0.4978	-0.0036	0.4985
	0.1921	0.3762	-0.0112	0.4955	-0.0010	0.4994
	-0.0202	0.4919	-0.0524	0.4791	-0.0247	0.4901
	0.4402	0.0701	-0.0449	0.4854	0.0028	0.3011
l	-0.8054	0.2103	-0.0181	0.4928	-0.0023	0.4991
	0.3174	0.6245	0.0172	0.5069	-0.0039	0.4984
	0.3564	0.6392	0.0094	0.5038	-0.0002	0.4999
	-0.2530	0.4001	0.0056	0.5023	0.0001	0.5000
	0.1694	0.5672	0.0015	0.5006	-0.0010	0.4996
	0.5323	0.7027	0.0292	0.5117	-0.0024	0.4991
	0.1074	0.5427	-0.0292	0.4883	-0.0062	0.4975
	0.3966	0.6542	0.0257	0.5103	0.0000	0.5000
	0.4198	0.6627	0.0111	0.5044	-0.0003	0.4999
~ 5	0.1148	0.5457	0.0181	0.5072	0.0007	0.5003
/ °)	0.0275	0.5110	-0.0217	0.4914	0.0018	0.5007
	0.2383	0.5942	0.0023	0.5009	-0.0016	0.4994
	0.1895	0.5751	0.0097	0.5039	0.0003	0.5001
	0.3384	0.6325	0.0107	0.5043	0.0020	0.5008
	-0.2167	0.4142	0.0096	0.5038	0.0009	0.5003
	-0.3862	0.3497	-0.2792	0.3900	0.0192	0.5077
	-0.7932	0.2138	0.0139	0.5055	-0.0020	0.4992
	0.1030	0.5410	-0.0021	0.4992	0.0014	0.5006
,	0.0131	0.5052	-0.0119	0.4953	0.0016	0.0006

Table 2: Test for false loans

Notes: see Table 1. T is defined in equation 4.

Table 3: Tests of similarity between EONIA and FEONIA

Variables: EONIA and FEONIA samples

Time period		[11/6/2014 -	9/9/2014]			[9/9/2014]	- 7/10/2014]
FEONIA Sample	Stand	lard algo	New pr	ocedure	Stand	lard algo	New p	orocedure
Test	Т	Р	Т	Р	Т	Р	Т	Р
Kolmogorov-Smirnov Ansari-Bradley Wilcoxon Kruskal-Wallis	0.4242 3.8962 -3.0197 9.1324	$\begin{array}{c} 7.748 \text{e-}06 \\ 9.7706 \text{e-}05 \\ 0.0025 \\ 0.0025 \end{array}$	$\begin{array}{c} 0.1818 \\ 2.2576 \\ -0.6576 \\ 0.4355 \end{array}$	$\begin{array}{c} 0.2025 \\ 0.0239 \\ 0.5107 \\ 0.5093 \end{array}$	$\begin{array}{c} 0.3500 \\ -4.0079 \\ 0.1758 \\ 0.0358 \end{array}$	$\begin{array}{c} 0.1349 \\ 6.1254 \text{e-} 05 \\ 0.8604 \\ 0.8498 \end{array}$	$0.1524 \\ -2.4914 \\ 1.0144 \\ 1.0567$	$\begin{array}{c} 0.2572 \\ 0.0127 \\ 0.3103 \\ 0.3039 \end{array}$

Notes: T is the test statistic, P is the p-value. The Kolmogorov-Smirnov is a nonparametric hypothesis test that evaluates the difference between the CDF. The Ansari-Bradley test is a nonparametric alternative to the two-sample F-test of equal variances. The Wilcoxon rank sum test is equivalent to the Mann-Whitney U-test. The Mann-Whitney U-test is a nonparametric test for the equality of population medians of two independent samples. The Kruskal-Wallis test is a nonparametric version of classical one-way ANOVA, and an extension of the Wilcoxon rank sum test to more than two groups. It compares the medians of the groups.

	Estimates	Tes	Test		
		Z	Р		
	(1)	(2)	(3)		
α	0.7143^{***} (0.0077)	-0.0147	0.494		
((0.0011)				
(0.0226***	0.0344	0.513		
	(0.0019) 0.0128^{***}	0.0585	0.523		
	(0.0045)	0.2772	0.647		
	(0.0031)	0.3773	0.047		
	0.0301^{***} (0.0049)	0.0115	0.5040		
	0.0386***	0.0855	0.534		
	0.0636***	0.0354	0.514		
	(0.0034) 0.0206^{***}	0.0900	0.5358		
	(0.0019) 0.0032	0.0590	0.523		
	(0.0020)	0.0000	0 509		
β s	(0.0032)	0.0223	0.508		
	(0.0316^{***})	0.0075	0.5030		
	0.0011 (0.0034)	0.0130	0.505		
	0.0113***	0.0522	0.5208		
	0.0155***	-0.2347	0.4072		
	(0.0032) 0.0327^{***}	0.0365	0.514		
	(0.0034) -0.0651***	0.0315	0.512		
	(0.0045)	0.0121	0.505		
	(0.0050)	0.0131	0.505.		
l	0.0932^{***} (0.0053)	0.1303	0.5518		
(-0.0312***	-0.0350	0 4860		
	(0.0018)	0.0000	0.100		
	(0.0018) -0.0434*** (0.0026)	-0.0360	0.4850		
	(0.0018) -0.0434*** (0.0026) -0.0075***	-0.0360 0.2225	0.485		
	$\begin{array}{c} (0.0018) \\ -0.0434^{***} \\ (0.0026) \\ -0.0075^{***} \\ (0.0028) \\ -0.0210^{***} \end{array}$	-0.0360 0.2225 -0.0115	0.4850		
	$\begin{array}{c} (0.0018) \\ -0.0434^{***} \\ (0.0026) \\ -0.0075^{***} \\ (0.0028) \\ -0.0210^{***} \\ (0.0051) \\ -0.0477^{***} \end{array}$	-0.0360 0.2225 -0.0115 -0.0729	0.4850 0.5880 0.4954 0.4709		
	$\begin{array}{c} (0.0018) \\ -0.0434^{***} \\ (0.0026) \\ -0.0075^{***} \\ (0.0028) \\ -0.0210^{***} \\ (0.0051) \\ -0.0477^{***} \\ (0.0016) \\ 0.0067^{**} \end{array}$	-0.0360 0.2225 -0.0115 -0.0729	0.485 0.588 0.495 0.470		
	$\begin{array}{c} (0.0018) \\ -0.0434^{***} \\ (0.0026) \\ -0.0075^{***} \\ (0.0028) \\ -0.0210^{***} \\ (0.0051) \\ -0.0477^{***} \\ (0.0016) \\ 0.0067^{**} \\ (0.0034) \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373	0.4850 0.5880 0.4954 0.4709 0.4709		
	$\begin{array}{c} (0.0018) \\ -0.0434^{***} \\ (0.0026) \\ -0.0075^{***} \\ (0.0028) \\ -0.0210^{***} \\ (0.0051) \\ -0.0477^{***} \\ (0.0016) \\ 0.0067^{**} \\ (0.0034) \\ -0.0406^{***} \\ (0.0022) \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633	0.485 0.588 0.495 0.470 0.485 0.474		
	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0016)\\ 0.0067^{**}\\ (0.0034)\\ -0.0406^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0020)\end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0636	0.485 0.588 0.495 0.470 0.485 0.474 0.474		
	$\begin{array}{c} (0.0018) \\ -0.0434^{***} \\ (0.0026) \\ -0.0075^{***} \\ (0.0028) \\ -0.0210^{***} \\ (0.0051) \\ -0.0477^{***} \\ (0.0016) \\ 0.0067^{**} \\ (0.0034) \\ -0.0406^{***} \\ (0.0022) \\ -0.0422^{***} \\ (0.0020) \\ -0.0242^{***} \\ (0.0027) \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0636 0.0127	0.485 0.588 0.495 0.470 0.485 0.474 0.474 0.474		
γs	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0016)\\ 0.0067^{**}\\ (0.0034)\\ -0.0406^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0020)\\ -0.0242^{***}\\ (0.0037)\\ 0.0059^{*}\\ \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0636 0.0127 -0.0035	0.4856 0.5886 0.495- 0.4709 0.4853 0.4744 0.4744 0.5053		
γs	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0016)\\ 0.0067^{**}\\ (0.0034)\\ -0.0406^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0037)\\ 0.0029^{*}\\ (0.0035)\\ -0.0258^{***} \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0633 -0.0636 0.0127 -0.0035 -0.0113	0.4850 0.5880 0.4955 0.4709 0.4855 0.4740 0.4855 0.4744 0.5055 0.4980 0.4980		
γs	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0016)\\ 0.0067^{**}\\ (0.0022)\\ -0.0422^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0023)\\ -0.0242^{***}\\ (0.0035)\\ -0.0258^{***}\\ (0.0026)\\ -0.0258^{***}\\ (0.0026)\\ -0.0200^{***} \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0636 0.0127 -0.0035 -0.0113 -0.0279	0.485 0.588 0.495 0.470 0.485 0.470 0.485 0.474 0.505 0.498 0.498 0.495		
γs	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0051)\\ -0.0467^{***}\\ (0.0034)\\ -0.0406^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0035)\\ -0.0242^{***}\\ (0.0035)\\ -0.0258^{***}\\ (0.0026)\\ -0.0200^{***}\\ (0.0030)\\ -0.0200^{**}\\ -0.0200^{**}\\ -0.0200^{**}\\ -0.0200^{*}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0633 -0.0636 0.0127 -0.0035 -0.0113 -0.0279	0.4850 0.5880 0.4955 0.4709 0.4855 0.4744 0.5055 0.4980 0.4980 0.4955 0.4889		
γs	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0016)\\ 0.0067^{**}\\ (0.0034)\\ -0.0406^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0020)\\ -0.0242^{***}\\ (0.0037)\\ 0.0059^{*}\\ (0.0035)\\ -0.0258^{***}\\ (0.0035)\\ -0.0200^{***}\\ (0.0030)\\ -0.0500^{***}\\ (0.0030) \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0633 -0.0636 0.0127 -0.0035 -0.0113 -0.0279 -0.0419	0.485 0.588 0.495 0.470 0.485 0.474 0.505 0.498 0.498 0.495 0.488 0.488		
γs	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0016)\\ 0.0067^{**}\\ (0.0034)\\ -0.0406^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0020)\\ -0.0242^{***}\\ (0.0037)\\ 0.0059^{*}\\ (0.0037)\\ -0.0258^{***}\\ (0.0026)\\ -0.0208^{***}\\ (0.0036)\\ -0.0200^{***}\\ (0.0030)\\ -0.0500^{***}\\ (0.0030)\\ -0.0500^{***}\\ (0.0035) \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0633 -0.0636 0.0127 -0.0035 -0.0113 -0.0279 -0.0419 0.2940	0.485 0.588 0.495 0.470 0.485 0.474 0.505 0.498 0.495 0.495 0.488 0.483 0.615		
γs	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0016)\\ 0.0067^{**}\\ (0.0034)\\ -0.0406^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0020)\\ -0.0242^{***}\\ (0.0035)\\ -0.0200^{***}\\ (0.0036)\\ -0.0200^{***}\\ (0.0036)\\ -0.0200^{***}\\ (0.0035)\\ -0.0284^{***}\\ (0.0035)\\ 0.2884^{***}\\ (0.0036)\\ -0.2844^{***}\\ (0.0036)\\ -0.2844^{***}\\ (0.0036)\\ -0.2844^{***}\\ (0.0036)\\ -0.2844^{***}\\ (0.0036)\\ -0.2844^{***}\\ (0.0036)\\ -0.2844^{***}\\ (0.0036)\\ -0.2844^{**}\\ -0.00440\\ -0.2844^{**}\\ -0.00440\\ -0.2844^{**}\\ -0.00440\\ -0.2844^{**}\\ -0.00440\\ -0.2844^{**}\\ -0.0040\\ -0.2844^{**}\\ -0.0040\\ -0.2844^{**}\\ -0.0040\\ -0.2844^{**}\\ -0.0040\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.2844^{**}\\ -0.004\\ -0.00$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0636 0.0127 -0.0035 -0.0113 -0.0279 -0.0419 0.2940 -0.0316	0.485 0.588 0.495 0.470 0.485 0.474 0.505 0.498 0.495 0.495 0.495 0.488 0.483 0.615 0.487		
γ s{	(0.0018) - 0.0434^{***} (0.0026) - 0.0075^{***} (0.0028) - 0.0210^{***} (0.0051) - 0.0477^{***} (0.0016) 0.0067^{**} (0.0022) - 0.0422^{***} (0.0022) - 0.0422^{***} (0.0022) - 0.0422^{***} (0.0037) - 0.0242^{***} (0.0037) - 0.0258^{***} (0.0035) - 0.0258^{***} (0.0028) - 0.0200^{***} (0.0030) - 0.0500^{***} (0.0035) - 0.0200^{***} (0.0035) - 0.0200^{***} (0.0035) - 0.0284^{***} (0.0035) - 0.2884^{***} (0.0044) 0.1417^{***}	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0636 0.0127 -0.0035 -0.0113 -0.0279 -0.0419 0.2940 -0.0316 -0.0284	0.4850 0.5880 0.4955 0.4709 0.4857 0.4740 0.4857 0.4980 0.4980 0.4883 0.4883 0.6150 0.4887 0.4887		
γs	$\begin{array}{c} (0.0018)\\ -0.0434^{***}\\ (0.0026)\\ -0.0075^{***}\\ (0.0028)\\ -0.0210^{***}\\ (0.0051)\\ -0.0477^{***}\\ (0.0051)\\ -0.0407^{***}\\ (0.0034)\\ -0.0406^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0022)\\ -0.0422^{***}\\ (0.0037)\\ -0.0242^{***}\\ (0.0037)\\ -0.0258^{***}\\ (0.0035)\\ -0.0258^{***}\\ (0.0035)\\ -0.0258^{***}\\ (0.0036)\\ -0.0200^{***}\\ (0.0030)\\ -0.0500^{***}\\ (0.0030)\\ -0.0190^{***}\\ (0.0044)\\ 0.1417^{***}\\ (0.0054)\\ -0.0074^{***}\\ \end{array}$	-0.0360 0.2225 -0.0115 -0.0729 -0.0373 -0.0633 -0.0636 0.0127 -0.0035 -0.0113 -0.0279 -0.0419 0.2940 -0.0316 -0.0284 -0.0131	0.4850 0.5880 0.4950 0.4709 0.4850 0.4744 0.5050 0.4980 0.4950 0.4950 0.4882 0.4883 0.6150 0.4874 0.4887 0.4887		

Table 4: Clean sample

Notes: see Table 1.

Figures

Figure 1: Time series of daily average rate detected by the Furfine algorithm and the EONIA rate. (19/05/2008 - 7/10/2014)



Notes: The violet vertical line traces the first sovereign debt crisis in April 2010, the black vertical line traces the second sovereign debt crisis in August 2011, the green line traces the ECB announcement of the Outright Monetary Transactions programme and the light blue line traces the signal rate change in July 2012 with the OD = 0%. The red lines traces the signal rate change in July 2012 with the OD = 0%. The red lines traces the signal rate change in July 2012 with the OD = 0%.

Figure 2: Time series of daily volume and value of loans detected by the Furfine algorithm. (19/05/2008 - 7/10/2014)



Notes: zero rates included by 11/06/2014, see Figure 1.

Figure 3: Empirical distribution functions of rates detected by the Furfine algorithm. (11/06/2014 - 7/10/2014)



Notes: Blue curves refer to maintenance periods with OD = -0.1% and ML = +0.4%, red curves refer to maintenance periods with OD = -0.2% and ML = +0.3%.





Notes: see Figure 3.

Figure 5: Time series of daily volume and value of loans returned by the econometric procedure. (19/05/2008 - 7/10/2014)



Notes: see Figure 1.





Notes: see Figure 3.



Figure 7: Empirical distribution functions of NEW FEONIA and EONIA. (11/06/2014 - 7/10/2014)

Notes: see Figure 3.

Figure 8: Robustness check - time series of daily volume and value of loans detected by the Furfine algorithm. (19/05/2008 - 7/10/2014)



Figure 9: Robustness check - Time series of daily volume and value of loans returned by the econometric procedure. (19/05/2008 - 7/10/2014)





Figure 10: Robustness check - e-MID frequency of zero interest loans. (01/04/2014 - 7/10/2014)

RECENTLY PUBLISHED "TEMI" (*)

- N. 1028 *The impact of CCPs' margin policies on repo markets*, by Arianna Miglietta, Cristina Picillo and Mario Pietrunti (September 2015).
- N. 1029 European structural funds during the crisis: evidence from Southern Italy, by Emanuele Ciani and Guido de Blasio (September 2015).
- N. 1030 Female employment and pre-kindergarten: on the uninteded effects of an Italian reform, by Francesca Carta and Lucia Rizzica (September 2015).
- N. 1031 The predictive content of business survey indicators: evidence from SIGE, by Tatiana Cesaroni and Stefano Iezzi (September 2015).
- N. 1032 Sovereign debt exposure and the bank lending channel: impact on credit supply and the real economy, by Margherita Bottero, Simone Lenzu and Filippo Mezzanotti (September 2015).
- N. 1033 *Does trend inflation make a difference?*, by Michele Loberto and Chiara Perricone (September 2015).
- N. 1034 Procyclicality of credit rating systems: how to manage it, by Tatiana Cesaroni (September 2015).
- N. 1035 *The time varying effect of oil price shocks on euro-area exports*, by Marianna Riggi and Fabrizio Venditti (September 2015).
- N. 1036 Domestic and international macroeconomic effects of the Eurosystem expanded asset purchase programme, by Pietro Cova, Patrizio Pagano and Massimiliano Pisani (September 2015).
- N. 1037 Deconstructing the gains from trade: selection of industries vs. reallocation of workers, by Stefano Bolatto and Massimo Sbracia (November 2015).
- N. 1038 Young adults living with their parents and the influence of peers, by Effrosyni Adamopoulou and Ezgi Kaya (November 2015).
- N. 1039 Shoe-leather costs in the euro area and the foreign demand for euro banknotes, by Alessandro Calza and Andrea Zaghini (November 2015).
- N. 1040 *The macroeconomic effects of low and falling inflation at the zero lower bound*, by Stefano Neri and Alessandro Notarpietro (November 2015).
- N. 1041 The use of fixed-term contracts and the (adverse) selection of public sector workers, by Lucia Rizzica (November 2015).
- N. 1042 Multitask agents and incentives: the case of teaching and research for university professors, by Marta De Philippis (November 2015).
- N. 1043 *Exposure to media and corruption perceptions*, by Lucia Rizzica and Marco Tonello (November 2015).
- N. 1044 *The supply side of household finance*, by Gabriele Foà, Leonardo Gambacorta, Luigi Guiso and Paolo Emilio Mistrulli (November 2015).
- N. 1045 *Optimal inflation weights in the euro area*, by by Daniela Bragoli, Massimiliano Rigon and Francesco Zanetti (January 2016).
- N. 1046 *Carry trades and exchange rate volatility: a TVAR approach*, by Alessio Anzuini and Francesca Brusa (January 2016).
- N. 1047 A new method for the correction of test scores manipulation by Santiago Pereda Fernández (January 2016).
- N. 1048 *Heterogeneous peer effects in education* by by Eleonora Patacchini, Edoardo Rainone and Yves Zenou (January 2016).
- N. 1049 *Debt maturity and the liquidity of secondary debt markets*, by Max Bruche and Anatoli Segura (January 2016).
- N. 1050 *Contagion and fire sales in banking networks*, by Sara Cecchetti, Marco Rocco and Laura Sigalotti (January 2016).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- A. MERCATANTI, A likelihood-based analysis for relaxing the exclusion restriction in randomized experiments with imperfect compliance, Australian and New Zealand Journal of Statistics, v. 55, 2, pp. 129-153, TD No. 683 (August 2008).
- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, v. 11, 2, pp. 433-465, **TD No. 709 (May 2009).**
- F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).**
- D. DOTTORI, S. I-LING and F. ESTEVAN, *Reshaping the schooling system: The role of immigration*, Journal of Economic Theory, v. 148, 5, pp. 2124-2149, **TD No. 726 (October 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian Selection*, Journal of International Economics, v. 89, 1, pp. 96-109, **TD No. 728 (October 2009).**
- L. MONTEFORTE and G. MORETTI, *Real-time forecasts of inflation: the role of financial variables*, Journal of Forecasting, v. 32, 1, pp. 51-61, **TD No. 767 (July 2010).**
- R. GIORDANO and P. TOMMASINO, *Public-sector efficiency and political culture*, FinanzArchiv, v. 69, 3, pp. 289-316, **TD No. 786 (January 2011).**
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, v. 59, pp. 212-227, TD No. 793 (February 2011).
- F. NUCCI and M. RIGGI, *Performance pay and changes in U.S. labor market dynamics*, Journal of Economic Dynamics and Control, v. 37, 12, pp. 2796-2813, **TD No. 800 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 805 (April 2011).**
- A. ACCETTURO e L. INFANTE, Skills or Culture? An analysis of the decision to work by immigrant women in Italy, IZA Journal of Migration, v. 2, 2, pp. 1-21, TD No. 815 (July 2011).
- A. DE SOCIO, *Squeezing liquidity in a "lemons market" or asking liquidity "on tap"*, Journal of Banking and Finance, v. 27, 5, pp. 1340-1358, **TD No. 819 (September 2011).**
- S. GOMES, P. JACQUINOT, M. MOHR and M. PISANI, Structural reforms and macroeconomic performance in the euro area countries: a model-based assessment, International Finance, v. 16, 1, pp. 23-44, TD No. 830 (October 2011).
- G. BARONE and G. DE BLASIO, *Electoral rules and voter turnout*, International Review of Law and Economics, v. 36, 1, pp. 25-35, **TD No. 833 (November 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, v. 11, 5, pp. 1032-1052, **TD No. 835 (November 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, in G. Gomel, D. Marconi, I. Musu, B. Quintieri (eds), The Chinese Economy: Recent Trends and Policy Issues, Springer-Verlag, Berlin, TD No. 838 (November 2011).
- A. ANZUINI, M. J. LOMBARDI and P. PAGANO, *The impact of monetary policy shocks on commodity prices*, International Journal of Central Banking, v. 9, 3, pp. 119-144, **TD No. 851 (February 2012).**
- R. GAMBACORTA and M. IANNARIO, *Measuring job satisfaction with CUB models*, Labour, v. 27, 2, pp. 198-224, **TD No. 852 (February 2012).**
- G. ASCARI and T. ROPELE, Disinflation effects in a medium-scale new keynesian model: money supply rule versus interest rate rule, European Economic Review, v. 61, pp. 77-100, TD No. 867 (April 2012)
- E. BERETTA and S. DEL PRETE, Banking consolidation and bank-firm credit relationships: the role of geographical features and relationship characteristics, Review of Economics and Institutions, v. 4, 3, pp. 1-46, TD No. 901 (February 2013).
- M. ANDINI, G. DE BLASIO, G. DURANTON and W. STRANGE, *Marshallian labor market pooling: evidence from Italy*, Regional Science and Urban Economics, v. 43, 6, pp.1008-1022, **TD No. 922 (July 2013).**
- G. SBRANA and A. SILVESTRINI, Forecasting aggregate demand: analytical comparison of top-down and bottom-up approaches in a multivariate exponential smoothing framework, International Journal of Production Economics, v. 146, 1, pp. 185-98, TD No. 929 (September 2013).
- A. FILIPPIN, C. V, FIORIO and E. VIVIANO, *The effect of tax enforcement on tax morale,* European Journal of Political Economy, v. 32, pp. 320-331, **TD No. 937 (October 2013).**

- G. M. TOMAT, *Revisiting poverty and welfare dominance*, Economia pubblica, v. 44, 2, 125-149, **TD No. 651** (December 2007).
- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, v.46, 4, pp. 693-713, **TD No. 730 (October 2009).**
- G. MICUCCI and P. ROSSI, *Il ruolo delle tecnologie di prestito nella ristrutturazione dei debiti delle imprese in crisi*, in A. Zazzaro (a cura di), Le banche e il credito alle imprese durante la crisi, Bologna, Il Mulino, **TD No. 763 (June 2010).**
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di politica economica, v. 105, 1, pp. 301-321, **TD No. 787 (January 2011).**
- R. BRONZINI and E. IACHINI, Are incentives for R&D effective? Evidence from a regression discontinuity approach, American Economic Journal : Economic Policy, v. 6, 4, pp. 100-134, TD No. 791 (February 2011).
- P. ANGELINI, S. NERI and F. PANETTA, *The interaction between capital requirements and monetary policy*, Journal of Money, Credit and Banking, v. 46, 6, pp. 1073-1112, **TD No. 801 (March 2011).**
- M. BRAGA, M. PACCAGNELLA and M. PELLIZZARI, *Evaluating students' evaluations of professors,* Economics of Education Review, v. 41, pp. 71-88, **TD No. 825 (October 2011).**
- M. FRANCESE and R. MARZIA, Is there Room for containing healthcare costs? An analysis of regional spending differentials in Italy, The European Journal of Health Economics, v. 15, 2, pp. 117-132, TD No. 828 (October 2011).
- L. GAMBACORTA and P. E. MISTRULLI, Bank heterogeneity and interest rate setting: what lessons have we learned since Lehman Brothers?, Journal of Money, Credit and Banking, v. 46, 4, pp. 753-778, TD No. 829 (October 2011).
- M. PERICOLI, *Real term structure and inflation compensation in the euro area*, International Journal of Central Banking, v. 10, 1, pp. 1-42, **TD No. 841 (January 2012).**
- E. GENNARI and G. MESSINA, How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data, International Tax and Public Finance, v. 21, 2, pp. 324-344, TD No. 844 (January 2012).
- V. DI GACINTO, M. GOMELLINI, G. MICUCCI and M. PAGNINI, *Mapping local productivity advantages in Italy: industrial districts, cities or both?*, Journal of Economic Geography, v. 14, pp. 365–394, TD No. 850 (January 2012).
- A. ACCETTURO, F. MANARESI, S. MOCETTI and E. OLIVIERI, Don't Stand so close to me: the urban impact of immigration, Regional Science and Urban Economics, v. 45, pp. 45-56, TD No. 866 (April 2012).
- M. PORQUEDDU and F. VENDITTI, Do food commodity prices have asymmetric effects on euro area inflation, Studies in Nonlinear Dynamics and Econometrics, v. 18, 4, pp. 419-443, TD No. 878 (September 2012).
- S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics, v. 76, 3, pp. 389-410, **TD No. 879 (September 2012).**
- F. D'AMURI and G. PERI, *Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession,* Journal of the European Economic Association, v. 12, 2, pp. 432-464, TD No. 886 (October 2012).
- M. TABOGA, What is a prime bank? A euribor-OIS spread perspective, International Finance, v. 17, 1, pp. 51-75, **TD No. 895 (January 2013).**
- G. CANNONE and D. FANTINO, *Evaluating the efficacy of european regional funds for R&D*, Rassegna italiana di valutazione, v. 58, pp. 165-196, **TD No. 902 (February 2013).**
- L. GAMBACORTA and F. M. SIGNORETTI, Should monetary policy lean against the wind? An analysis based on a DSGE model with banking, Journal of Economic Dynamics and Control, v. 43, pp. 146-74, **TD No. 921 (July 2013).**
- M. BARIGOZZI, CONTI A.M. and M. LUCIANI, Do euro area countries respond asymmetrically to the common monetary policy?, Oxford Bulletin of Economics and Statistics, v. 76, 5, pp. 693-714, TD No. 923 (July 2013).
- U. ALBERTAZZI and M. BOTTERO, Foreign bank lending: evidence from the global financial crisis, Journal of International Economics, v. 92, 1, pp. 22-35, TD No. 926 (July 2013).

- R. DE BONIS and A. SILVESTRINI, *The Italian financial cycle: 1861-2011*, Cliometrica, v.8, 3, pp. 301-334, **TD No. 936 (October 2013).**
- G. BARONE and S. MOCETTI, *Natural disasters, growth and institutions: a tale of two earthquakes,* Journal of Urban Economics, v. 84, pp. 52-66, **TD No. 949 (January 2014).**
- D. PIANESELLI and A. ZAGHINI, *The cost of firms' debt financing and the global financial crisis*, Finance Research Letters, v. 11, 2, pp. 74-83, **TD No. 950 (February 2014).**
- J. LI and G. ZINNA, *On bank credit risk: sytemic or bank-specific? Evidence from the US and UK*, Journal of Financial and Quantitative Analysis, v. 49, 5/6, pp. 1403-1442, **TD No. 951 (February 2015).**
- A. ZAGHINI, Bank bonds: size, systemic relevance and the sovereign, International Finance, v. 17, 2, pp. 161-183, **TD No. 966 (July 2014).**
- G. SBRANA and A. SILVESTRINI, *Random switching exponential smoothing and inventory forecasting,* International Journal of Production Economics, v. 156, 1, pp. 283-294, **TD No. 971 (October 2014).**
- M. SILVIA, Does issuing equity help R&D activity? Evidence from unlisted Italian high-tech manufacturing firms, Economics of Innovation and New Technology, v. 23, 8, pp. 825-854, TD No. 978 (October 2014).

2015

- M. BUGAMELLI, S. FABIANI and E. SETTE, The age of the dragon: the effect of imports from China on firmlevel prices, Journal of Money, Credit and Banking, v. 47, 6, pp. 1091-1118, TD No. 737 (January 2010).
- R. BRONZINI, The effects of extensive and intensive margins of FDI on domestic employment: microeconomic evidence from Italy, B.E. Journal of Economic Analysis & Policy, v. 15, 4, pp. 2079-2109, TD No. 769 (July 2010).
- G. BULLIGAN, M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors,* International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- A. CIARLONE, *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- D. FANTINO, A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, v. 1, 2, pp. 219-251, TD No. 884 (October 2012).
- D. DEPALO, R. GIORDANO and E. PAPAPETROU, Public-private wage differentials in euro area countries: evidence from quantile decomposition analysis, Empirical Economics, v. 49, 3, pp. 985-1115, TD No. 907 (April 2013).
- G. BARONE and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916 (June 2013).**
- P. ALESSANDRI and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- M. TANELI and B. OHL, *Information acquisition and learning from prices over the business cycle*, Journal of Economic Theory, 158 B, pp. 585–633, **TD No. 946 (January 2014).**
- R. AABERGE and A. BRANDOLINI, *Multidimensional poverty and inequality*, in A. B. Atkinson and F. Bourguignon (eds.), Handbook of Income Distribution, Volume 2A, Amsterdam, Elsevier, TD No. 976 (October 2014).
- V. CUCINIELLO and F. M. SIGNORETTI, *Large banks,loan rate markup and monetary policy*, International Journal of Central Banking, v. 11, 3, pp. 141-177, **TD No. 987 (November 2014).**
- M. FRATZSCHER, D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**
- A. NOTARPIETRO and S. SIVIERO, Optimal monetary policy rules and house prices: the role of financial frictions, Journal of Money, Credit and Banking, v. 47, S1, pp. 383-410, TD No. 993 (November 2014).
- R. ANTONIETTI, R. BRONZINI and G. CAINELLI, *Inward greenfield FDI and innovation*, Economia e Politica Industriale, v. 42, 1, pp. 93-116, **TD No. 1006 (March 2015).**
- T. CESARONI, *Procyclicality of credit rating systems: how to manage it,* Journal of Economics and Business, v. 82. pp. 62-83, **TD No. 1034 (October 2015).**
- M. RIGGI and F. VENDITTI, *The time varying effect of oil price shocks on euro-area exports,* Journal of Economic Dynamics and Control, v. 59, pp. 75-94, **TD No. 1035 (October 2015).**

- G. DE BLASIO, D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds*, Industrial and Corporate Change, **TD No. 792 (February 2011).**
- A. DI CESARE, A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, **TD No. 831 (October 2011).**
- E. BONACCORSI DI PATTI and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, TD No. 848 (February 2012).
- M. MARCELLINO, M. PORQUEDDU and F. VENDITTI, Short-Term GDP Forecasting with a mixed frequency dynamic factor model with stochastic volatility, Journal of Business & Economic Statistics, **TD No. 896 (January 2013).**
- M. ANDINI and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, **TD No. 915 (June 2013).**
- F BRIPI, *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, **TD No. 932 (September 2013).**
- G. ALBANESE, G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, **TD No. 955 (March 2014).**
- A. L. MANCINI, C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, **TD No. 958 (April 2014).**
- R. BRONZINI and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, **TD No.** 960 (April 2014).
- L. BURLON, *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory, **TD** No. 961 (April 2014).
- L. BURLON and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, **TD No. 963 (June 2014).**
- A. BRANDOLINI and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, **TD No. 965 (July 2015).**
- G. ZINNA, Price pressures on UK real rates: an empirical investigation, Review of Finance, TD No. 968 (July 2014).
- A. CALZA and A. ZAGHINI, Shoe-leather costs in the euro area and the foreign demand for euro banknotes, International Journal of Central Banking, **TD No. 1039 (December 2015).**